Author's response to reviews

Title: The effect of posterior subtenon methylprednisolone acetate in the refractory diabetic macular edema: a prospective nonrandomized interventional case series

Authors:

Alireza Javadzadeh (javadzadehalireza@yahoo.com)

Version: 4 Date: 24 March 2006

Author's response to reviews:

Dear BioMed Central Editorial Team,

In response to your Email 21 Mar 2006 concerning the manuscript "The effect of posterior subtenon methylprednisolone acetate in the refractory macular edema: prospective nonrandomized interventional case series". I would like also to thank the reviewer for their useful and scientific comments. The comments were taken into consideration as follows.

1- Title and abstract format were revised.
2- Author list was corrected.
3- References were revised.

Yours sincerely
Alireza Javadzadeh